Author's response to reviews

Title: Nurse anaesthetists- and OR nurses` adherence to WHO's Surgical Safety Checklist -a grounded theory study

Authors:

Hilde V Waehle 1 (hilde.valen.waehle@helse-bergen.no)
Arvid S Haugen 1,2 (arvid.haugen@helse-bergen.no)
Eirik Softeland 1 (eirik.softeland@helse-bergen.no)
Esther Hjälmhult 3 (Esther.Hjaelmhult@hib.no)

Version: 5 Date: 10 July 2012

Author's response to reviews:

Revisions in the manuscript:
Adjusting team involvement:
A grounded theory study of challenges in utilizing a surgical safety checklist as experienced by nurses in the operating room.

-Justification for the design using grounded theory, and the research question related to the findings.
-Justification of the combination of focus group interviews- and individual interview, in conjunction with the observation in OR.
-Information of the data collection, the samling process, observation in the OR, and achievement of data saturation.
-Elements related to truth and validity in grounded theory, the criteria of fit, work, relevance and modifiability have been added.
-Improved introduction to each of the quotes from the interviews.
-The entire discussion part has been rewritten, and relevant literature has been added.
-Title of the manuscript has been changed.
-Manuscript has been proof read by a native speaker.